Author's response to reviews

Title: Partial complex regional pain syndrome type 1 of the hand: retrospective study of 16 cases and literature review.

Authors:

michel konzelmann (michel.konzelmann@crr-suva.ch)
françois luthi (francois.luthi@crr-suva.ch)

Version: 2 Date: 30 July 2012

Author's response to reviews: see over
Dear Dr Majithia,

Please find enclosed the manuscript entitled “Partial complex regional pain syndrome type 1 of the hand: retrospective study of 16 cases and literature review”, by Michel Konzelmann and François Luthi submitted for publication in BMC Neurology.

This manuscript contains original results and has not been submitted anywhere else for publication. It’s the largest well documented series of partial CRPS type 1 of the hand with long term follow-up. This is the first to use the recently validated criteria of CRPS (Budapest’s criteria). Even relatively rare, this particular form of CRPS is important to know for clinicians.

We declare no competing interest in relation with this work.

We made the changes concerning the ethics and the consent statements. In the manuscript we include these changes in the Methods section line 69 to 72.

I am looking forward for your response.

Sincerely,

Michel Konzelmann, MD